Author's response to reviews

Title: Impact of a task focusing strategy on perceived stress levels and performance during a simulated cardiopulmonary resuscitation: A randomized controlled trial

Authors:

Sabina Hunziker (sabinahunziker@gmail.com)
Simona Pagani (simona.pagani@unibas.ch)
Katrin Fasler (kfasler@gmx.ch)
Franziska Tschan (Franziska.Tschan@unine.ch)
Norbert Semmer (norbert.semmer@psy.unibe.ch)
Stephan Marsch (smarsch@uhbs.ch)

Version: 2 Date: 15 July 2012

Author's response to reviews:

Dear Editor

I have now registered the trial at clinicaltrials.gov (http://clinicaltrials.gov/) as requested; the record will be released in 2-5 working days

Warm regards

Sabina Hunziker